Mauna Loa Sol ar Observatory Cbserver’'s Log
Sat Feb 17 17:35:12 GV 1996
Year: 96 Doy: 048
oserver: koon consol e Feb 17 17:33

WEATHER COMMENT: Sat Feb 17 17:35:18 GMI 1996
Thick cirrostratus and cirrocunul us overcast, w nd=30 nph fromthe WsW
temp=33 F. Cbserving is unlikely for today.

LOML COMVENT: Sat Feb 17 18:27:40 GMI 1996

Found bot h tapes popped out of drives, programwas still running and

i nstrument head was pointing at the sun. Pushed L00374 back into drive #0
and renmoved L0O0373, cleaned tape drive #1 (lights were blinking), then
installed LO0O375 in drive #1. Cycled power to the conputer and to each
drive then restarted program

WEATHER COMMENT: Sat Feb 17 19:35: 00 GMI 1996
Clouds are getting thicker, no observing today.

TAPE:
* ok ok ok k

LOML: LO0374 in drive #0

Sat Feb 17 19:35:45 GMI 1996: Fil emark



